Primitives for the Manipulation of General Subdivisions and the Computation of Voronoi Diagrams

Leo J. Guibas and Jorge Stolfi Xerox PARC and Stanford University

Abstract: We discuss the following problem: given n points in the plane (the "sites"), and an arbitrary query point q, find the site that is closest to q. This problem can be solved by constructing the Voronoi diagram of the given sites, and then locating the query point in one of its regions. We give two algorithms, one that constructs the Voronoi diagram in $O(n \lg n)$ time, and another that inserts a new site in O(n) time. Both are based on the use of the Voronoi dual, the Delaunay triangulation, and are simple enough to be of practical value. The simplicity of both algorithms can be attributed to the separation of the geometrical and topological aspects of the problem, and to the use of two simple but powerful primitives, a geometric predicate and an operator for manipulating the topology of the diagram. The topology is represented by a new data structure for generalized diagrams, that is embeddings of graphs in two-dimensional manifolds. This structure represents simultaneously an embedding, its dual, and its mirror-image. Furthermore, just two operators are sufficient for building and modifying arbitrary diagrams.

O. Introduction

One of the fundamental data structures of computational geometry is the Voronoi diagram. This diagram arises from consideration of the following natural problem. Let n points in the plane be given, called sites. We wish to preprocess them into a data structure, so that given a new query point q, we can efficiently locate the nearest neighbor of q among the sites. The n sites in fact partition the plane into a collection of n regions, each associated with one of the sites. If region P is associated with site p, then P is the locus of all points in the plane closer to p than to any of the other n - 1 sites. This partition is known as the Voronoi diagram (or the Dirichlet, or Thiessen, tesselation) determined by the given sites.

The closest site problem is can therefore be solved by constructing the Voronoi diagram, and then locating the query point in it. Using the currently best available algorithms, the Voronoi diagram of n points can be computed in $O(n \lg n)$ time and stored in O(n) space; these bounds have been shown to be optimal in the worst case [Sh]. Once we have the Voronoi diagram, we can construct in linear further time a structure with which we can do point location in a planar subdivision in $O(\lg n)$ time [Ki1].

The work of Jorge Stolfi, who is on leave from the University of São Paulo (São Paulo, Brazil) was partially supported by a grant from Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq).

Permission to copy without fee all or part of this material is granted provided that the copies are not made or distributed for direct commercial advantage, the ACM copyright notice and the title of the publication and its date appear, and notice is given that copying is by permission of the Association for Computing Machinery. To copy otherwise, or to republish, requires a fee and/or specific permission. Shamos [Sh] first pointed out that the Voronoi diagram can be used as a powerful tool to give efficient algorithms for a wide variety of other geometric problems. Given the Voronoi, we can compute in linear time the closest pair of sites, or the closest neighbor of each site, or the Euclidean minimum spanning tree of the *n* sites, or the largest point-free circle with center inside their convex hull, etc. Several of these problems are known to require $\Omega(n \lg n)$ time in the worst case, so these Voronoi-based algorithms are asymptotically optimal.

The complexity of the $O(n \lg n)$ Voronoi algorithms that can be found in the literature is a serious barrier to their widespread utilization. To the authors' knowledge, they so far have never been used in any of the significant practical applications of closest-point problems in statistics, operations research, geography, and other areas. In every case the authors of those programs chose to use asymptotically slower $O(n^2)$ algorithms, which were much simpler to code and almost certainly faster for the ranges of interest in *n*. Furthermore, the presentation of Voronoi algorithms in the literature has often been insufficiently precise. Authors typically confine themselves only to a very high-level description of their algorithms. As with many geometric problems, difficulties can arise in the implementation when degeneracies occur, as for example when three of the given points happen to be cocircular.

In this paper we present a novel way of looking at the standard Voronoi computation techniques, such as the divide and conquer [SH] and incremental [GS] methods, that results in Voronoi algorithms that are very concise and substantially easier to read, implement and verify. One of the main reasons for this simplicity is that we work with the duals of the Voronoi diagrams, which are known as *Delaunay triangulations*, rather than with the diagrams themselves. The other major reason is the clean separation that we are able to make between topological and geometrical aspects of the problem. In sections 6 through 10 we show that the hardest part of constructing a Voronoi diagram or Delaunay triangulation is the determination of its topological structure, that is, the incidence relations between vertices, edges, and faces. Once the topological properties of the diagram are known, the geometrical ones (coordinates, angles, lengths, etc.) can be computed in time linear in the size of the diagram.

Our algorithms are built using essentially two primitives: a geometric predicate, and a topological operator for manipulating the structure of the diagrams. The geometrical primitive, that we call the *InCircle* test, encapsulates the essential geometric information that determines the topological structure of the Voronoi diagram, and is a powerful tool not only in the coding of the algorithms but also in proving their correctness. As evidence for its importance, we show that it possesses many interesting properties, and can be defined in a number of equivalent ways.

The topological structure of a Voronoi or Delaunay diagram is equivalent to that of a particular embedding of some undirected graph in the Euclidean plane. We have found it convenient to consider such diagrams as being drawn on the sphere rather than on the plane; topologically that is equivalent to augmenting the Euclidean plane by a dummy *point at infinity*. This allows us to represent such things as infinite edges and faces in the same way as their finite counterparts. In sections 1 through 5 we will establish the mathematical properties of such embeddings, define a notation for talking about them, and describe a data structure for their representation.

It turns out that the data structure we propose is general enough to allow the representation of undirected graphs embedded in arbitrary two-dimensional manifolds. In fact, it may be seen as a variant of the "winged edge" representation for polyhedral surfaces [Ba]. We show that a single topological operator, which we call *Splice*, together with a single primitive for the creation of isolated edges, is sufficient for the construction and modification of arbitrary diagrams. Our data structure has the ability to represent simultaneously and uniformly both the primal, the dual, and the mirror-image diagrams, and to switch arbitrarily from one of these domains to another, in constant time. Finally, the design of the data structure enables us to manipulate its geometrical and topological parameters independently of each other. As it will become clear in the sequel, these properties have the effect of producing programs that are at once simple, elegant, efficient from a practical point of view, and asymptotically optimal in time and space.

Since this paper is quite long, some guidance to the forthcoming sections may be advisable. Section 1 introduces the concept of a simple subdivision of a manifold and discusses some of the conventions we adopt as compared to the extant literature. Section 2 presents the very important ideas of the dual of a subdivision, and the edge algebra associated with a subdivision. The edge algebra is a combinatorial structure on the edges of the subdivision that we claim captures all the topological information associated with the subdivision. Section 3 is more technical and may be omitted on a first reading. It formalizes and then proves the above claim about edge algebras by showing that isomorphism of edge algebras is equivalent to topological homeomorphism between the corresponding subdivisions. In section 4 we present a computer representation for an edge algebra, which is our quad edge data structure. Section 5 introduces the topological primitives that we use to manipulate this structure and discusses their properties and implementation. Section 6 tailors these primitives to the application on hand, namely the Delaunay/Voronoi computation. Section 7 reviews some properties of the Voronoi/Delauanay subdivision and section 8 presents our main geometric primitive for their computation, the InCircle test and its properties. Section 9 presents in detail and proves correct a divide and conquer algorithm for Voronoi computations, and section 10 discusses incremental techniques.

1. Subdivisions

In this section we will give a precise definition for the informal concept of an embedding of an undirected graph on a surface. Special instances of this concept are sometimes referred to as a subdivision of the plane, a generalized polyhedron, a two-dimensional diagram, or by other similar names. They have been extensively discussed in the solid modeling literature of computer graphics [Ba, MS]. We want a definition that accurately reflects the topological properties one would intuitively expect of such embeddings (for instance, that every edge is on the boundary of two faces, every face is bounded by a closed chain of edges and vertices, every vertex is surrounded by a cyclical sequence of faces and edges, and so forth) and at the same time is as general as possible and leads to a clean theory and data structure.

We assume the reader is familiar with a few basic concepts of pointset topology, such as topological space, continuity, and homeomomorphism [IK]. Two subsets A and B of a topological space M are said to be *separable* if some neighborhood of A is disjoint from some neighborhood of B; otherwise, they are said to be *incident* on each other. A *line* of M is a subspace of M homeomorphic to the open interval $B^1 = (0 \ 1)$ of the real line. A *disk* of M is a subspace homeomorphic to the open circle of unit radius $B^2 = \{x \in \mathbb{R}^2 : |x| < 1\}$. Recall that a *two-dimensional manifold* is a topological space with the property that every point has an open neighborhood which is a disk (all manifolds in this paper will be two-dimensional).

- Definition 1.1. A subdivision of a manifold M is a partition S of M into three finite collections of disjoint parts, the vertices, the edges, and the faces (denoted respectively by VS, $\mathcal{E}S$, and $\mathcal{F}S$), with the following properties:
 - S1. Every vertex is a point of M.
 - S2. Every edge is a line of M.
 - S3. Every face is a disk of M.
 - S4. The boundary of every face is a closed path of edges and vertices.

The vertices, edges, and faces of a subdivision are called its *elements*. Figure 1.1 shows some examples of subdivisions.



Figure 1.1. Examples of subdivisions.

Condition S4 needs some explanation. We will denote by \overline{B}^2 the closed circle of unit radius, and by S^1 its circumference. Let us define a simple path in S^1 as a partition of S^1 into a finite sequence of

isolated points and open arcs. The precise meaning of S4 is then the following: for every face F there is a simple path π in S^1 and a continuous mapping ϕ_F from \overline{B}^2 onto the closure of F that (i) maps homeomorphically B^2 onto F, (ii) maps homeomorphically each arc of π into an edge of S, and (iii) maps each isolated point of π to a vertex of S.

Condition S4 has a number of important implications. Clearly the points and arcs of π must alternate as we go around S^1 ; if α is the arc between two consecutive points a and b of π , then its image $\phi_F(\alpha)$ is an edge incident to the points $\phi_F(a)$ and $\phi_F(b)$. Therefore, the images of the elements of π , taken in the order in which they occur around S^1 , constitute a closed, connected path π_F of edges and vertices of S, whose union is the boundary of F. Notice that the bounding path π_F need not be simple, since ϕ_F may take two or more distinct arcs or points of π to the same element of S. Therefore the closure of a face may not be homeomorphic to a disk, as figure 1.1 shows.

Since it is impossible to cover a disk with only a finite number of edges and vertices, every edge and every vertex in a subdivision of a manifold must be incident to some face. We conclude that every edge is entircly contained in the boundary of some face, and that it is incident to two (not necessarily distinct) vertices of S. These vertices are called the *endpoints* of the edge; if they are the same, then the edge is a *loop*, and its closure is homeomorphic to the circle S^1 .

Since every element of S is in the closure of some face, and since the closed disk \overline{B}^2 is compact, the manifold M is the union of a finite number of compact sets — and therefore is itself compact. In fact, condition S4 can be replaced by the requirement that M be compact, that the edges be pairwise separable, and that every vertex is incident to some edge. Furthermore, every compact manifold has a subdivision. We will not attempt to prove these statements, since they are too technical for the scope of this paper.

Informally speaking, a compact two-dimensional manifold is a surface that closes upon itself, has no boundary, and in which every infinite sequence has an accumulation point. The sphere, the torus, and the projective plane are such manifolds; the disk, the line segment, the whole plane, and the Möbius strip are not. The compactness condition is not as restrictive as it may seem; any manifold can be made compact by adding a dummy "point at infinity" that is by definition an accumulation point of all sequences with no other accumulation points. This operation transforms the Euclidean plane R^2 into the extended plane, which is homeomorphic to the sphere.

1.1. Equivalence and connectivity

Definition 1.2. Let S and S' be two subdivisions of the manifolds M and M'. An *isomorphism* from S to S' is a homeomorphism of M onto M' that maps each element of S onto an element of S'. When such a mapping exists, we say that S and S' are *equivalent*, and we write $S \sim S'$.

Such an isomorphism will perforce map vertices into vertices, faces into faces, edges into edges, and will preserve the incidence relationships among them. A *topological property* of subdivisions is a property that is invariant under equivalence. Our goal will be to develop a computer representation that fully captures all topological properties of subdivisions.

The collection of all edges and vertices of a subdivision S constitutes an undirected graph, the graph of S. The graphs of two equivalent subdivisions S and S' are obviously isomorphic. The converse is not always true: if S and S' have isomorphic graphs, it doesn't follow that they are equivalent, or that M and M' are homeomorphic. Figure 1.2 shows an example. Note that the subdivisions are not equivalent even though there also is a one-to-one correspondence between the faces of S and S' with the property that corresponding faces are incident to corresponding edges and vertices. This example shows that the *set* of edges and vertices on the boundary of a face is not enough information to characterize its relationship to the rest of the manifold.



Figure 1.2. Two subdivisions with isomorphic graphs that are not equivalent.

This fact is the main source of complexity in the theoretical treatment of subdivisions, notably in the proof that our data structure is a consistent representation of a general subdivision. It is possible to define subdivisions in such a way that their topological structure is completely determined by that of their graphs. For example, if the manifold is restricted to be a sphere, and the graph is triply connected [Har], then the subdivision is determined up to equivalence. However, any set of conditions strong enough to achieve this goal would probably outlaw "degeneracies" such as loops, multiple edges with the same endpoints, faces with nonsimple boundaries, and so forth. Subdivisions with such degeneracies are much more common than it may seem: they inevitably arise as intermediate objects in the transformation of a "well-behaved" subdivision into another. An even stronger reason for adopting our liberal definition is that it leads to more flexible data structures and simpler atomic operations with weaker preconditions.

On the other hand, we depart from the common solid modeling tradition by insisting that every face be a simple disk, without "handles" or "holes", even though the whole manifold is allowed to have arbitrary connectivity. The main reason for this requirement is to enable a clean and unambiguous definition of the dual subdivision (see subsection 2.2). One important consequence of this restriction is stated below:

Theorem 1.1. The graph of a simple subdivision is connected iff the manifold is connected.

Proof: Since every face is incident to some edge, if the graph is connected then the whole manifold is too. Now assume the the graph is not connected, but the manifold is. Since the faces are pairwise separable, and their addition to the graph makes it connected, some face is incident to two distinct components of the graph. By condition S4 the boundary of that face is connected, a contradiction. \Box

Therefore, the connected components of the manifold are in one-toone correspondence with the connected components of the underlying graph.

2. The edge algebra of a subdivision

In this section we will develop a convenient notation for describing relationships among edges of a subdivision, and a mathematical framework that will justify the choice of our data structure. We will develop first the theory and representation for arbitrary compact manifolds, and then we will show that certain important simplifications can be made in the particular case when the manifold is orientable. For many applications, including the computation of Voronoi diagrams, the only relevant manifold will be the extended plane.

2.1 Basic edge functions

On any disk D of a manifold there are exactly two ways of defining a local "clockwise" sense of rotation; these are called the two possible orientations of D. An oriented element of a subdivision P is an element x of P together with an orientation of a disk containing x. There are also exactly two consistent ways of defining a linear order among the points of a line ℓ ; each of these orderings is called a *direction along* ℓ . A *directed edge* of a subdivision P is an edge of P together with a direction along it. Since directions and orientations can be chosen independently, for every edge of a subdivision there are four directed, oriented edges. Observe that this is true even if the edge is a loop, or is incident twice to the same face of P.

For any oriented and directed edge e we can define unambiguously its vertex of origin, eOrg, its destination, eDest, its left face, eLeft, and its right face, eRight. We define also the flipped version eFlipof an edge e as being the same unoriented edge taken with opposite orientation and same direction, as well as the symmetric of e, eSym, as being the same undirected edge with the opposite direction but the same orientation as e. We can picture the orientation and direction of an edge e as a small bug sitting on the surface over the midpoint of the edge and facing along it. Then the operation eSym corresponds to the bug making a half turn on the same spot, and eFlip corresponds to the bug hanging upside down from the other side of the surface, but still at the same point of the edge and facing the same way.

The elements eOrg, eLeft, eRight, and eDest are taken by definition with the orientation that agrees locally with that of e. More precisely, the orientation of eOrg agrees with that of some initial segment of e, and that of eDest agrees with some final segment of e. Note that for some loops eOrg and eDest may have opposite orientations, in spite of being the same (unoriented) vertex. Similarly, the orientation of eLeft agrees with e along the "left margin" of e, and that of eRightagrees along its "right margin". If e is a bridge, it may be the case that eLeft and eRight have different orientations, in spite of being the same (unoriented) face. By extending our previous notation, we will denote by VS, ES and FS the sets of directed and oriented elements of a subdivision S. In the rest of this section, unless otherwise specified, all subdivision elements are assumed to be oriented, and directed if edges.

A sufficiently small disk containing the vertex v = eOrg, it can be mapped homeomorphically onto the unit disk B^2 in such a way that v is mapped to the origin, and the intersection of D with every edge incident to v is a ray of B^2 . Traversing the boundary of Din the counterclockwise direction (as defined by the orientation of v) establishes a cyclical ordering of those edges. If each edge is oriented so as to agree with v, and directed away from D, we obtain what is called the *ring of edges out of v*. We can define the *next edge with same origin, eOnext*, as the one immediately following e (counterclockwise) in this ring (see figure 2.1). Note that if e is a loop it will occur twice in the ring of it (either e Sym or e SymFlip) will occur once each: since the manifold around v is like a disk, e will occur only once in each circuit, and we will never encounter eFlip.

Similarly, given an edge e we define the next counterclockwise edge with same left face, denoted by e Lnext, as being the first edge we encounter

after *e* when moving along the boundary of the face F = eLeft, in the counterclockwise sense as determined by the orientation of *F*. The edge *eLnext* is oriented and directed so that *eLnextLeft* = *F* (including orientation). The successive images of *e* under *Lnext* give precisely the edges of the bounding path π_F of condition S4 (in one of the two possible orders). As in the case of *Onext*, the edge *e* appears exactly once in this list, and either *eSym* or *eFlip* (but not *eSymFlip*) may appear once.



Figure 2.1. The ring of edges out of a vertex.

2.2. Duality

The dual of a planar graph G can be defined intuitively as a graph G^* obtained from G by interchanging vertices and faces while preserving the incidence relationships. The definition below extends this intuitive concept to arbitrary subdivisions:

Definition 2.1. Two subdivisions S and S^* are said to be *dual* of each other if for every directed and oriented edge e of either subdivision there is another edge e Dual of the other such that

D1. (e Dual) Dual = eD2. (e Sym) Dual = (e Dual) SymD3. (e Flip) Dual = (e Dual) Flip SymD4. $(e Lnext) Dual = (e Dual) Onext^{-1}$

Equation D4 states that moving counterclockwise around the left face of e in one subdivision is the same as moving clockwise around the origin of (eDual) in the other subdivision. To see why, note that the edges on the boundary of the face F = eLeft, in counterclockwise order, are $(eLnext, eLnext^2, ..., eLnext^m = e)$ for some $m \ge 1$. This path maps through Dual to the sequence $((eDual)Onext^{-1}, (eDual)Onext^{-2}, ..., (eDual)Onext^{-m} = eDual)$ of all edges coming out of the vertex v = (eDual)Org of S^* , in clockwise order around v.

We can therefore extend *Dual* to vertices and faces of the two subdivisions, by defining (*eLeft*) *Dual* = (*eDual*) *Org* and (*eOrg*) *Dual* = (*eDual*) *Left*. Equations D2 and D3 imply that any two edges that differ only in orientation and direction will be mapped to two versions of the same undirected edge. Combining this with the preceding argument we conclude that *Dual* establishes a correspondence between *ES* and *ES**, between *VS* and *FS**, and between *FS* and *VS**, such that incident elements of *S* correspond to incident elements of *S**, and vice-versa. It follows that two vertices of one subdivision are connected by an edge whenever (and as many times as) the corresponding faces of the other are incident to a common edge. So, in the particular case when *S* and *S** are subdivisions of the sphere, the graphs of *S* and *S** are duals of each other in the sense of graph theory. Figure 2.2 shows a subdivision of the extended plane (solid lines) superimposed on its dual (dotted lines). Note that the two subdivisions of figure 2.2 have the property that each undirected edge of one meets (and crosses) only the corresponding dual edge of the other, and that each vertex of one is in the corresponding dual face of the other. When this happens, we say that S and S^* are *strict duals* of each other. In that case, the dual of an oriented and directed edge e is the edge of the dual subdivision that crosses e from left to right, but taken with orientation *opposite* to that of e. That is, the dual subdivision should be looked from the other side of the manifold, or the manifold should be turned inside out. This reflects the correspondence between counterclockwise traversal of eLeft to clockwise traversal of (eDual)Org.



Figure 2.2. A subdivision of the extended plane (solid lines) and a strict dual (dashed lines).

This implicit "flipping" of the manifold is unavoidable if S and S^* are superimposed as strict duals and we insist that *Dual* be its own inverse. It has the serious drawback of making the calculus of the edge functions much less intuitive. It is therefore preferable to relate the two dual subdivisions by means of the function

which maps ES to ES^* without this implicit "flipping". The edge *e Rot* is called the *rotated* version of *e*; it is the undirected dual of *e*, directed from *e Right* to *e Left*, and oriented so that moving counterclockwise around the right face of *e* corresponds to moving counterclockwise around the origin of *e Rot*. If the two subdivisions are superimposed as strict duals, like in figure 2.2, then we may say that *e Rot* is *e* "rotated 90° counterclockwise" around the crossing point. In fact, the only reason for not defining duality in terms of *Rot* (rather than *Dual*) is that it fails short of being its own inverse: (*e Rot*) Rot gives *e Sym* instead of *e*.

2.3. Properties of edge functions

The functions Flip, Rot, and Onext satisfy the following properties:

E1. $eRot^4 = e$ E2. eRotOnextRotOnext = eE3. $eRot^2 \neq e$ E4. $e \in ES$ iff $eRot \in ES^*$ E5. $e \in ES$ iff $eOnext \in ES$

- F1. $eFlip^2 = e$
- F2. eFlipOnextFlipOnext = e
- F3. $eFlipOnext^n \neq e$ for any n
- F4. eFlipRotFlipRot = e
- F5. $e \in ES$ iff $e Flip \in ES$

A number of useful properties can be deduced from these, as for example

$$eFlip = eFlip$$

 $eSym = eRot^2$
 $eRot^{-1} = eRot^3$
 $= eFlipRotFlip$
 $eDual = eFlipRot$
 $eOnext^{-1} = eRotOnextRot$
 $= eFlipOnextFlip$,

and so forth. For added convenience in talking about subdivisions, we introduce some derived functions. By analogy with *eLnext* and *eOnext*, for a given *e* we define the *next edge with same right face*. *eRnext*, and with same destination, *eDnext*, as the first edges that we encounter when moving counterclockwise from *e* around *eRight* and *eDest*, respectively. These functions satisfy also the following equations:

$$eLnext = eRot^{-1}OnextRot$$

 $eRnext = eRotOnextRot^{-1}$
 $eDnext = eSymOnextSym$

The orientation and direction of the returned edges is defined so that eLnextLeft = eLeft, eRnextRight = eRight, and eDnextDest = eDest. Note that eRnextDest = eOrg, rather than vice-versa. By moving *clockwise* around a fixed endpoint or face, we get the inverse functions, defined by

$$e Oprev = e Onext^{-1} = e Rot Onext Rot$$

$$e Lprev = e Lnext^{-1} = e Onext Sym$$

$$e Rprev = e Rnext^{-1} = e SymOnext$$

$$e Dprev = e Dnext^{-1} = e Rot^{-1} Onext Rot^{-1}$$

It is important to notice that every function defined so far (except Flip) can be expressed as the composition of a constant number of *Rot* and *Onext* operations, independently of the size or complexity of the subdivision. Figure 2.3 illustrates these various functions.



Figure 2.3. The edge functions.

2.4. Edge algebras

efinition 2.1 An edge algebra is an abstract algebra $\{E, E^*, Onext, Rot, Flip\}$, where E and E^* are arbitrary finite sets, and Onext, Rot, and Flip are functions on E and E^* satisfying properties F1-F5 and E1-E5. Peninten

facts of S^{*}. These orbits are joined by *Hip* in pairs of opposite orientation. We can therefore define eOr in an edge algebra as the orbit of the under Ontext, and similarly $eLeff = eRo^{-1}O_{2x}$, $eRight = eOin (ar eOr)_{2x}$ is the same as $eO_{2x}Hip$ (the same as $eO_{2x}Hip$ (the same as $eO_{2x}Hip$ (the same as $eO_{2x}Hip$) (t The axioms imply that Rot is a bijection from E to E^* and from E^* to E_* . Note frypt and Onext each define premutations sturg on <math>E and E^* separately. The orbits of Onext in E are in one-to-one E and E^* separately. The orbits of Onext in E are in one-to-one conrespondence with the vertices of S, and therefore also with the

Edge algebras are a purely combinatorial abstraction of subdivisions, in the same way that undirected graphs are an abstraction of the subdivision graphs. In the next section we will defend the position that all topological properties of a subdivision S are accurately contrust that all topological properties of a subdivision S are accurately contrust by an edge algebra. As a consequence this algebra, which is a finite object, can be used as a computer representation of S. An edge algebra before, this allows us to express all our edge functions in terms of only three basic primitives. *Flip. Rot.* and *Onext.* Other advantages of this represents simultaneously a pair of dual subdivisions; as we remarked primal/dual representation will be encountered later on, and we will j j see that they are obtained at a negligible cost in storage and

3. From edge algebras back to subdivisions

by its seacuted edge algebra. The concepts and theorems developed in this searchistic edge algebra. The concepts and theorems developed in this search are searching for showing the consistent and completeness of the data structure but are not used in the rest of the paper, so the reader whose interest is mostly practical can skip to section 4. The major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be to show that a general studitivision S can be fully major facts will be the study of t In the present section we will show that the topological properties of an arbitrary subdivision S are accurately and unambiguously represented urn is closely related to the edge algebra of S.

3.1. Completions

Definition 3.1. Let S and Σ be subdivisions of a manifold M. We that Σ is a completion of S if it is a refinement of S obtained by adding one vertex c_s on each edge e and one vertex v_F in each face F, and then connecting v_F by new edges to every vertex (old or new) on the boundary of F. The vertices of Σ are called *primal, crossing,* or *dual* depending on whether help lie on vertices, edges or faces of Σ , they are denoted by VE. (Σ), and V'E, respectively. Fvery edge of S is spilt by its crossing vertex in two *primal links* of Σ ; the new edges added in each face are realled *dual* links? if they connect a *dual vertex* to a crossing point, and sfew *links* if they connect a *dual vertex* to a crossing point, and sfew *links* if they connect a *dual vertex* to a crossing point. and sfew *links* if they connect a *dual vertex* to a crossing point. and sfew *links* if they connect a *dual vertex* to a crussing These links are denoted $L\Sigma$, L, Σ , and $K\Sigma$ in that order. Figure 3.1 shows a completion of a subdivision of the extended plane.

Definition 3.1 must be understood appropriately in the case of a face F whose bounding path π_F is not simple. If π_F passes k times through a vertex or crossing point p, then p is to be connected to v_F by exactly k new links, and their order around v_F should be the same as the order of the crossings around π_F . To describe this process precisely, let ϕ_F be any continuous function from \overline{B}^2 to the closure of F that establishes

condition S4. Let $\pi = (u_1, \alpha_1, u_2, \alpha_2, ..., u_{n_1}, \alpha_{n_2}, u_{n+1} = u_1)$ be the path in the circle S¹ that is mapped to $\pi \neq by \phi_{\pi^{-1}}$ in each are α_1 there is a point c_1 hat is mapped to the recosing vertex of the edge $\phi_{\pi}(\alpha_1)$. Take $\phi_{\pi}((\alpha_1))$ to be the dual vertex v_2 : connect in \overline{B} the origin (0, 0) to each u_1 and to each c_1 by a straight line segment, and and skew links for the face F. Note that the restriction of faces to simple disks is essential for a simple and unambiguous definition of completion ğ



and plane. showing primal links (solid), dual links (dashed) skew links (dotted). Figure 3.1. A completion on the extended

one refinement which is a completion. Every face of Σ consists of distinct. An important consequence is that the closure of each face distinct. An important consequence is that the closure of each face is homesmorphic to (not just the continuous image of) the sector of Ξ^{*} homesmorphic to a dask. In fact, the closure of a face of Σ is homesmorphic to a plast. In fact, the closure of a face of Σ is noncompopite to a plast. In fact, the closure of a sector of the s the definition, it is clear that every subdivision has at least From

than one link connecting any given pair of vertices. But it has no loops. Every crossing vertex c_{*} is incident to exactly four links, two primal and two dual, and to four distinct triangles. The vertex c_{*} and these eight tenents constitute a dist of M that contains the cdge 4. It can be seen also that, given a primal link ℓ and a dual link ℓ' that are incident to the same (crossing) vertex, there is exactly one triangle that It is also apparent from the definition that every edge of Σ has two distinct endpoints, and is incident to exactly two triangles (which may or may not lie in the same face of S). A completion may have more is incident to both L and L.

to be an integral part of the description of Σ_{-} so S is uniquely determined by it. We call S the prime additional subfigured of Σ_{-} denote by Σ_{-} in the same spirit, we say that two completions Σ_{+} and Σ_{-} are given the same spirit, we say that two completions Σ_{+} and Σ_{-} are or V_{-} provement of Σ_{+} to V_{-} provement of Σ_{-} to V_{-} and the same spirit, and Σ_{-} and the same spirit, we say that two completions of a data or V_{-} is on the most of Σ_{-} and Σ_{+} to V_{-} such an homeonorphism will clearly take $C\Sigma_{+}$, $L\Sigma_{+}$, $L\Sigma_{+}$, and $L\Sigma_{+}$ to V_{-} , and $L\Sigma_{+}$ to V_{-} and $L\Sigma_{+}$ to V_{-} and $L\Sigma_{+}$ to V_{-} and $L\Sigma_{+}$ to V_{-} and $L\Sigma_{+}$ to V_{-} . We consider the distinction between primal. dual and crossing vertices

3.2. Existence of duals and algebras

not only on S riself, but also on the choice of a dual subdivision S^{*}, and of the function Dual (or Rot) that connects the two. The first part of our theoretical justification is the proof that such S^{*} and Dual As it was defined, the edge algebra of a subdivision S seems to depend always exist, and that the edge functions of S and S^{*} satisfy E1-E3 and F1-F5.

Let Σ be a completion on a manifold M. For every crossing c_{i} of Σ , define the *ball* of the tunoriented and undirectorledge e of SD as the set $e^{-\frac{1}{2}} d_{i} - 1(c_{i} - 1)(c_{i}$, where t_{i}, t_{2} are the two dual links incident to c_{i} . Dranue by $t^{*} \Sigma$ the set of all seth objects. Define the *dual* F_{i}^{*} of a primal vertex v as the union of $\{v\}$ and all elements of Σ incident to v. Let $\mathcal{F}^*\Sigma$ be the set of all those objects.

Lemma 3.1. The triplet $S^*\Sigma = (V^*\Sigma, \mathcal{E}^*\Sigma, \mathcal{F}^*\Sigma)$ is a subdivision of N.

Proof: Besides v itself, the dual F_v^* of a vertex v contains only triangles, primal links and skew links incident to v. Each link of F_v^* is incident to exactly two distinct triangles of F_v^* , and conversely each of triangle and each t_i is incident to ξ_i and to ξ_{i+1} . Each such sequence plus v is a disk containing v; since M is a manifold, there can be only one is incident to two distinct links of F_{u}^{*} , one primal and one stew. Therefore, these links and triangles can be arranged in one or more sequences (without repetitions) $(L_1, L_2, L_2, L_3, \dots, L_n, L_n+1 = L_1)$ where the t, are triangles. the L_1 are alternately primal and skew links, such disk.

We conclude that F_v^* is a disk of M. Furthermore, it is clear that we can construct a continuous instruction ϕ from the closed ball onto the closenc F_v that esablishes condition S4. Since a transfer or primal time closure F_v indicates to two distinct primal vertices, the efferenses of 7.2 are pairwise disjoint. Clearly the elements of $\mathcal{E}^*\Sigma$ are lines of M that are pairwise disjoint, and also disjoint from the members of $\mathcal{F}^*\Sigma$ and $\mathcal{V}^*\Sigma$. Therefore $\mathcal{S}^*\Sigma$ is a subdivision of M.

Lefinition 3.1. Let Σ be a completion. Let Rot be the function from $ESD \cup RSY$ is not isself default as follows: for every deg $e \in ESD$. Let Rot be the dual edge e^* of S^*D , directed so as to cross e from right to left and ordened so as to agree with the orientation of e at the crossing point. Similarly, for each element $e \in RS^*D$ is terkor be the edge of SD of which e is the dual, directed and oriented according to the same rules with respect to e. The *standard* edge algebra of D is by definition AD = (ESE, ES* E, Onext, Rot, Hip)

Theorem 3.2. The standard edge algebra $A\Sigma$ of any completion Σ satisfies axioms EI-ES and FI-FS, and $S^*\Sigma$ is a (strict) dual of $S\Sigma$.

represented unanthiguously by a pair of links (e., e.), where e_{α} is the origin half of e. and e. is the dual (or primal) link of Σ that is incident to the crossing vertex of e and lies to its right. Conversely, any pair (i, p_i , p_i) adjacent links (one primal and one dual) corresponds to a unique edge of BSD or $BS^{-}\Sigma$. Proof: Each oriented and directed edge e of $ES\Sigma$ (or $ES^*\Sigma$) can be

For any link pair (x, y) of this kind there is a unique triangle T of Σ incident to z and y, and a unique triangle T sharing a stew link with T. Let scall the opposite of the pair (x, y) the link pair $\{r, s\}$ sich that r and a are on the boundary of T' and are of the same same strip (primal/dual) as z and y, respectively. Let A' denote the link pair $\{r, b\}$ (primal/dual) as z and y, respectively. Let z' denote the link of same sort (primal/dual) as z and incident to the same crossing.

According to this notation, we have (a, b) Flp = (a, b'). (a, b) Rot = (b, a'). and (a, b) Onext = (x, y) where (x, y) is opposite to (a, b').

Now it is easy to check that the algebra AE satisfies EJ-E5 and FJ-F5. For example, let (x, y) be the opposite of (b, a); then (a, b) is the opposite of (y, x), and we have (a, b) Rol Onexi Rol Onexi = (b, a') Onexi Rol Onexi = (x, y) Rol Onext= (y, x') Onext

= (a, b)

and so forth. The function e *Dual* = e*FlipRot* satisfied DI-DA, since these conditions can be proved from E1-FS and F1-FS. We conclude that SY and S'Y are (strict) duals of each other. \Box

For any subdivision S there is a completion Σ such that $S = S\Sigma$, and therefore a dual S*E and a valid edge algebra $A\Sigma$ that describes S (and S^{*}Σ).

3.2. Equivalence and isomorphism

The second part of our argument shows that the edge algebra of a subdivision is determined up to isomorphism, and conversely the subdivision of a edge algebra is unique up to equivalence. **Theorem 3.3.** Let A_i (i = 1, 2) be an edge algebra for a pair of dual subdivisions S_i and S_i^* . If S_1 is equivalent to S_2 , then A_1 and A_2 are isomorphic algebras

Proof: Let $A_i = (ES_i, ES_i, Onext_i, Roi,, Flip,)$, and let η be the homeomorphism between the manifolds of S_1 and S_2 that establishes their equivalence. An orientation or direction for an element of S_1 tween ES₁ and ES₂. From the definition of Onext we can conclude that $\eta(eOnext_1) = \eta(eOnext_2$ for all $e \in ES_1$; the same holds for ing element in S_2 , and so η is also a one-to-one correspondence bedetermines via η a unique orientation or direction for the correspond-Lnext and Flip. Let us now define the function ξ from $ES_1 \cup ES_1$ to $ES_2 \cup ES_2^*$ as

if $\epsilon \in ES_1$, $\binom{n(eRot_1^{-1})Rot_2}{1}$ if $e \in ES_1^*$. 4(e) (e) =

Clearly ξ is one-un-one. for Rat, is one-to-one from ES, to ES, Let us now show that $\{[eOnex_i]\} = \xi[e]Onex_i$. If $e \in ES$ the proof is trivial. If $e \in ES_1$, then $eOnex_i \in ES_1$, and $\xi(eOnext_1) = \eta(eOnext_1 Rot_1^{-1}) Rot_2$

 $= \eta(eRoi_1^{-1}Lprev_1Roi_1Roi_1^{-1})Roi_2$

= $\eta(eRoi_1^{-1})Lprev_2Roi_2$ (since $eRoi_1^{-1} \in ES$) = n(e Rot ⁻¹ Lprev₁) Rot₂

 $=\eta(eRoi_1^{-1})Roi_2Roi_2^{-1}Lprev_2Roi_2$ = {(e)0nex12

using The proof for $\xi(eRip_1) = \xi(e)Rip_2$ is entirely similar, usin $eRip_t = eRoi_t^{-1}Rip_1Roi_1$, and finally $\xi(eRoi_1) = \xi(e)Roi_2$ irvival. We say that two completions are similar if there is an isomorphism of the graph of Σ_1 to that of Σ_2 that takes primal vertices to primal vertices and dual vertices to dual vertices.

edge algebras AL, and AL2 are isomorphic algebras then Σ_1 and Σ_2 are similar Lemma 3.4. Let Σ_1 and Σ_2 be two completions If their

Proof: For any completion Σ , we stabilish onc-to-one mappings between certain subsets of oriextue and adirected edges of the the shear and vertices of Σ in the following way. To each primal links, dual links, dual link and following way. To each primal link of Σ there corresponds a

unique pair of primal (or dual) elements of $A\Sigma$ of the form {e, e, effip}; these elements are the directed and oriented edges of $S\Sigma$ (or $S^*\Sigma$) uses remains are volume to consist or the set of Σ there of which I is the "oright" half. To each primal vertex of Σ there corresponds an orbit of $\Delta\Sigma$ under I more and I is the set of the firm $e^{O_L} = O_L =$

We also associate each skew link of Σ to a set of the form

[e, e Flip, e Rol⁻¹, e Rol⁻¹ Flip, J, f Flip, f Roi, f Roi Flip}

traingles of Σ incident to s_i each incident also to a primal and to a dual link. We take s_i to be the union of the four subsets of $\Delta\Sigma$ that correspond to those four links. It is easy to therk that these subsets have the form above, and that s is incident to a primal or dual vertex of Σ if and only if an element of s' increases the orbit corresponding where f = eOnext, in the following way. There are exactly two to that vertex. Conversely, every set of the form above determines i unique skew link by this rule.

The isomorphism between $\Delta\Sigma_1$ and $\Delta\Sigma_2$ maps those representative subsets of $\Delta\Sigma_1$ to subsets of $\Delta\Sigma_2$ having the same form, and therefore it stabilises a one-correspondence ξ between the primal (or dual) inits and vertices of Σ_1 and hose of Σ_2 . Since intersecting subsets are mapped to intersecting subsets, ξ preserves incidence. We conclude Σ_1 and Σ_2 are similar. Lemma 3.5. If two completions Σ_1 and Σ_2 are similar, then they are equivalent.

homeomorphism η between the manifolds of the two completions that evaluations their equivalence. First we define η on the vectors of Σ_1 examines their equivalence. First we define η on the vectors of Σ_1 as being the same as ξ . For every that r of Σ_1 with endpoints up and v, we can always find an homeomorphism η , from the closure $\eta(y) = n/\mu \eta$ of all points μ of r. clearly, η is an homeomorphism $\eta(y) = n/\mu \eta$ for all points μ of r. Let ξ be the isomorphism between the graphs of Σ_i and it establishes their similarity. We will construct from it an that establishes their similarity. of the graph of Σ_1 onto that of Σ_2 . Proof: ភឹ

gives also a one-to-one correspondence between their triangles that preserves incidence. For each pair of corresponding triangles T and T there is a homeomorphism r_F of T mote T there is a submemorphism r_F of T mote T. This follows readily from the fact that both chosanes are homeencorphic to closed disks. So η and all η_T constitute a finite collection of continuous maps of closed subsets of M into M^2 , with the property that any two of them agree in the instruction of their domains. Their union η^2 is therefore a continuous map from M into M. Since any pair of adjacent links of which one is primal and the other dual determines a unique triangle, the similarity of the two completions Clearly, η^{*} is one-to-one and onto, so it is an homeomorphism construction, it maps elements of Σ_1 to elements of Σ_2 .

Lemma 3.5. If two completions Σ_1 and Σ_2 are equivalent, then so are SE1 and SE2.

Proof: Each face of SZ, is the union of a dual vertex and all elements of Σ_i that are incident to it. Each edge of SZ, is the union of a crossing and all (two) primal links of Σ_i incident to it. The homeomorphism neidence and the primal/dual character of links and vertices, so it maps elements of $S\Sigma_1$ to $S\Sigma_2$, establishing their equivalence. establishes the equivalence of the two completions preserves incidence and the chat

Theorem 3.6. Let A_1 and A_2 be edge algebras for two subdivisions S_1 and S_2 . If A_1 and A_2 are isomorphic, then S_1 and S_2 are equivalent.

Proof: Let Σ_1 and Σ_2 be any two completions of S_1 and S_2 . By theorem 31, we have $A_1 \sim A\Sigma_1$ and therefore $A\Sigma_1 \sim A\Sigma_2$, and therefore $A\Sigma_1 \sim A\Sigma_2$. Then by termma 3.4 Σ_1 and Σ_2 are equivalent; by the mma 3.1 the same is true of S_1 and S_2 .

Therefore, the topological structure of a subdivision is completely and uniquety characeterized by its edge algebra. Theorems 33 and 36 also imply that all completions of a subdivision are equivalent, and that two subdivisions are equivalent if and only if their duals are equivalent. Therefore, the dual of a simple subdivision is unique up to equivalence.

3.3. Realizability of algebras

To conclude our thoretical justification, we will show that every edge algebra corresponds to a subbivisoit of some analold. This fact is of great practical importance. for it guarantees that any modification to the data structure that haves axioms EI-ES and FI-FS invariant corresponds to a valid operation on manifolds.

Theorem 3.9. Every edge algebra can be realized by some subdivision.

Proof: Let $A = (B, E^*, Flip, Roy, Onexr)$ be an edge algebra. We will prove this by by constructing a completion Σ such that $A\Sigma$ is isomorphic to A. The manifold of Σ is constructed by taking a collection of disjoint closed triangles, that will become the triangles of a completion, and "pasting" their edges together as specified by A.

unordered pairs $\{e,e|f|g\}$ where $e \in E$. Similarly, let U^{*} denote the unoriented edges of E^{*}. We define a corner of the algebra as being a pair of unoriented edges of the form $\{\{e,e|f|g\}\}$, $\{e,e|f|g\}$, $\{e,e|f|g\}$, where e is an edge. Notice that there are |E| distinct corner in the algebra and that verty nonriented dee belongs to acardy we corners. Let 7 be a notherion of |E| disjoint closed triangles on the plane, each triangle 7, associated to a unique and distinct corner r of the Let then U be the set of all unoriented edges of A, that is, the set of all Label the three vertices of each triangle with the symbols V. algebra. ы ы For each unoriented edge $u \in U$, take the two conners r and s to which us belongs, and factify binoremorphically the VS addes of the two triangles T_r and T_r (matching V with V and E with E). That common side minus its two endpoints is the *primal link* corresponding to u. In the same manner. for every $u \in U^*$ are the two correst r and s increating u^* , and identify the FE sides of T_r and T_r ; the correston side will become the *dual link* corresponding to u^* .

Finally, for every corner $r = \{\{e, efip\}, \{eRoi, eRoi Fip\}\}$ there is easily non exposite corner $s = \{\{f_i, Fip\}, \{fRoi, Forlip\}\}$ such that f = eRoiOnersi and $e = fRoiOnersi. Identify the VF sides of <math>T_i$ and T_i . Call the same segment a vertex/gor link.

phic to a disk. Every side of every trangle is joined with exactly one side of a diskinet transitie, so a point on a link also has a disk the neighborhood. Now consider a vertex to of some transpe, and all other points that have been identified with it; they have all the some abort Clearly any point interior to a triangle has a neighborhood homeomornuction. An E type vertex v belongs to exactly four triangles by consti

{{eRot^k, eRot^k Flip}, {eRot^k Rot, eRot^k Rot Flip}} corresponding to the corners

for $0 \le k < 4$ and some edge e. Each triangle is pasted to the next one by a primal or dual link incident at v, so as to form a quadrilateral

with center v. A V or F type vertex v is common to 2n triangles (for {{ek, ek Flip}, {ek Rol, ek Rol Flip}} and come $n \ge 1$) corresponding to the corners

 $(\mathbf{e}, \mathbf{r}, f) Oprev = (\mathbf{e}[\mathbf{r} + 1 - f] \cdot \mathbf{Next}) Rol^{1-f} Flip^{f},$

and so forth.

 $(\mathbf{e}, \mathbf{r}, f)Sym = (\mathbf{e}, \mathbf{r} + 2, f)$ $(\mathbf{e}, \mathbf{r}, f)Rot^{-1} = (\mathbf{e}, \mathbf{r} + 3 + 2f, f)$

follows also that

where $e_k = eOnext^k$ for some edge e and $0 \le k < n$. These triangles are pasted alternately by vertex-face links and primal or dual {{ex Flip, ex}, {ex Flip Roi, ex Flip Roi Flip}},

links, so as to form a 2n-sided polygon around v. In all cases, the vertex v has a disk-like neighborhood.

We conclude that the triangles T pasted as above constitute a manifold. The links, the triangle interiors, and the identified vertices obviously define a completion Σ of this manifold, and $A\Sigma$ is isomorphic to $A \square$

The quad edge data structure 4

We represent a subdivision *S* (and simultaneously a dual subdivision *S*¹) by means of the quad angle data rencure, which is a natural computer implementation of the corresponding edge algebra. The edges of the algebra can be partitioned in groups of eight: each group consists of the four oriented and directed versions of an undirected edge 05 puts the town versions (is taket edge). The group containing a particular edge is therefore the ohtil of e under the subdiphon generated by *Rot* and *Fig.* To build the data structure, we select arbitrarily a cononical representative in cach group. Then any cdge e can be written as RRof Flipf', where $r \in \{0, 1, 2, 3\}$, $f \in \{0, 1\}$, and $\overline{\epsilon}$ is the canonical representative of the group to which e belongs.

The group of edges containing e is represented in the data structure by one *edge rector* (a) divided into four parts (a) through (a). The other and the regression of the edge 7.8.0°. Set figure 4.1a. A generic edge e = R R or Firg' is represented by the triplet (a,r, f), called an *dge reference*. We may think of this triplet as a pointer to the "Quarter-record" of Figure 1.7 thus a bit, f that tells whether we should took at it from "bow" of from "below".

Each part e(r) of an edge record contains two fields, Data and Wart. The Data field is used to hold geometrical and other non-topological information about the edge $\bar{e}Roi'$. This field meither affects nor is topological operations that we will describe, so its contents and format is entirely dependent on the application. affected by the

The Next field of \bullet [r] contains a reference to the edge *RAC Onext*. Given an arbitrary edge reference (\bullet , r, f), the three basic edge functions *Rot*. *Flip*, and *Onext* are given by the formulas

Ξ (\mathbf{e}, r, f) Onext = $(\mathbf{e} [r + f]$. Mext) Rot^f Flip^f (\mathbf{e}, r, f) Rol = $(\mathbf{e}, r + 1 + 2f, f)$ = (e, r, *j* + 1) (e, r, f) Flip

where the r and f components are computed modulo 4 and modulo respectively. In the first expression above, note that r + 1 + 2f

is congruent modulo 4 to r + 1 if f = 0, and r - 1 if f = 1; this construction to saying that rotating a 0° contraction the constant of the manifold, is the same as rotating it 0° bockwase as seen from the other side. Similarly, the third expression implies

þ (e, r, 0) *Onext* = e[r + f] . **Hex**t,

(c) The data structure for the subdivision (b). i.e., the moving counterclock wise around a vertex is the same as moving clockwise on the other side of the manifold. From these formulas it

Figure 4.1. The quad-edge data structure.

= (e, r, 0) Oprev Flip

 $(\mathbf{e}, r, 1)$ Onext = $(\mathbf{e} [r + 1]$. Next) Rot Flip

that

= (e, r, 0) Rol Onext Rol Flip

227

The record of an arbitrary edge belongs to four circular lists, cor-responding to its two regionism and its quade edge arbitrarians a portion of a subhivision and its quade edge arsurence. We may think of each record as belonging to four circular lists, corresponding to the vortices and two faces intenden to the edge. Note however that un traverse those lists we have to use to *Constri* function, not just the float, politiers. Consider for example the situation depicted in the set politiers.

figure 4.2, where the canonical representative of edge a has orientation

opposite to that of the others.

_ S

1



standard way to refer to it is to specify one of its outgoing edges. This has the added advantage of specifying a reference point on its ether may whether the frequently necessary when using the vertex as a parameter to topological operations. Similarly, the standard way of referention to a component of the edge structure is by ping once of the two dual subdivisions, and a "starting place" and "starting power of the two dual subdivisions, and a "starting place" and "starting direction" on it. Therefore a subdivision referred to by the cdge e can be "instantaneously" transformed into its dual by taking e Rot. quad edge data structure contains no separate records for vertices or faces; a vertex is implicitly defined as a ring of edges, and the



canonical repre-Figure 4.2. An Onext ring with can scattaives on both sides of the manifold.

4.1. Simplifications for orientable manifolds

we are going to discuss all manifolds to be handled are orientable. This means we can assign a specific orientation to each edge, wrtex and face of the subdivision so that any two incident clements have compatible orientations. Therefore the elements of the edge algebra can be partitioned in no sexts each focked under *Roi* and *Oraxi*, and each the image of the other under *Flip*. Then we don't need the *f* bit in edge references and the formulas simplify to In many applications, including the Voronoi/IDclaunay algorithms that

(e, r) Rot⁻¹ = (e, r + 3)
(e, r) Oprev = (e[r + 1] . Next) Rot, (e, r)*Onext ==* e[r] .Next = (o, r + 1) (e, r) Rol (e, r) Sym

und so forth.

edge algebra" that has only Onext and Sym as the primitive operators. Then we can get Data: Larve and Rarve in constant time, but not their invests. However, this may be adquarts for some applications. We save two pointers (and perihaps two data fields) in each edge We can represent a simple subdivision (without its dual) by a "simple record. Note that this optimization cannot be used with Flip.

4.2 Additional comments on the data structure

storage space required by the quad edge data structure, including pata fields, is $|EP| \times (8 \text{ record pointers} + 12 \text{ bits})$. The Ë ë

simplification for orientable manifolds reduces those 12 bits to 8. This comparts favorably with the winged-edge representation [fla] and with the Muller-Preparata variant [MP]. Indeed, all three representations use essentially the same pointers: each edge is connected to the four "immediately adjacent" ones (*Onest, Oprev, Dnext, Dprev*), and the four Data fields of our stneture may be seen as corresponding to the vence and face links of theirs.

since these usually come in pairs whose members are "dual" of each other. As an illustration of the flexibility of the quad edge structure consider the problem of constructing a diagram which is a cube joined of an occathedron: we can construct uso cubes (calling wire the same procedure) and join one to the dual of the other. Compared with the two versions mentioned above, the quad edge data surveiue has the advantage of allowing uniform access to the dual an mirror-image subdivisions. As we shall see, this capability allows us to cut in half the number of primitive and derived operations, us to cut in half the number of primitive and derived operations.

a straightforward programming exercise, given an auxiliary stack of size O(12E) and a bouldon mark bit on each directed deg (Rn). With a few more hits per edge, we can do away with the stack which filler the filler more algorithms can be used to enumerate the writtes of the studiotion, the sense of visiting stacky one dge unto every venter. If we take the dual studivision, we get an edge out of every venter. If we take the dual studivision, we get an edge out of every venter. If we take the dual studivision, we get an edge out of every venter. If we take the dual studivision, we get an edge out of every venter. The systematic enumeration of all edges in a (connected) subdivision is number of edges. Recall also that from Euler's relation it follows that number of vertices, edges, and faces of a subdivision are linearly å

Basic topological operators

ശ്

Perhaps the main advantage of the quad edge data structure is that the construction and modification of arbitrary diagrams can be effected by as few as two basic topological operators, in contrast to the half-dozen or more required by the previous versions [Br, MS].

The first operator is denoted by $e \leftarrow$ MakeEdge[]. It takes no parameters, and returns an edge e of a newly-created data structure representing a subdivision of the sphere (see fig. 5.1).



Figure 5.1. The result of MakeEdge.

Apart from orientation and direction, e will be the only edge of the subdivision, and will not be a loop; we have $eOrg \neq eDett, eLg/l = eRight, e. Inerci = eSixm, and e Oneta = eSixm, and e Oneta = eOrev = e. To construct a loop, we may use <math>e \rightarrow \text{lakeldge}[1, \text{Rot}$; then we will have $eOrg = eLg/l \neq eRight, eLnet = eRneta = e, and eOneta = eOrev = eSim.$

The second operator is denoted by Splite(a, b) and takes as parameters two edges a and b, returning no value. This operation affects the two edge rings a Org and b Org, and, independently, the two edge rings a Lgl and b Lgl. In each case,

· if the two rings are distinct, Sp11ce will combine them into one;

if the two are exactly the same ring. Splice will break it in two separate pieces:

if the two are the same ring taken with opposite orientations.
 Splice will *Flip* (and reverse the order) of a segment of that ring.

The parameters a and b determine the place where the edge rings will be cut and joined. For the rings a $0/r_{\rm g}$ and b $0/r_{\rm g}$, the cuts will occur immediately place a and b (in countercluctwise order); for the rings a L_2/d_1 and b L_2/d_1 , the cut will occur immediately before a for and b R_0 . Figure 3.2a illustrates this process for one of the simplest cases when a and have the same origin and distinct left faces. In this case Splitce (at. b) splits the remnon origin of a and b in two sparate vertices, and joins their left faces. If the origins are distinct and the left faces are the same, the effect will be precisely the opposite: the vertices are joined and the left faces are split, Indeed, splitce is its own inverse: if we perform Splitce [a, b] twice in a row we will get heard the same and the sime vertices of a now we will get hear the same addivision.



20rg = b0rg, a Left ≠ b Left



 $aOrg \neq bOrg, a Left = b Left$

Figure 5.2a. The effect of Splice: trading a vertex for a face.

Figure 5.2b illustrates the effect of Splite [a, b] in the case where a and b have distinct toff faces and distinct origins. In this case, Splice will either join two components in a single one, or add an extra "handk". u the matiloid, depending on whether a and b are in the same component or not. The quad-edge data structure the manifold. There seems to be no general way of doing this at a bounded cost per operation, on the other hand, in many applications this problem is trivial or statightforward, so it is best to solve this problem independently for each case. Figure 5.2b also illustrates the case when both left faces and ongins are distinct. to keep track automatically of the components and connectivity of the manifold. There seems to be no general way of doing this at a contains no mechanism to distinguish between these two cases, or



a Omexiko and $\beta = \delta Omexiko$, basically all we have to do is to interchange the values of d Omexiko, busically all we have to do is vith $\beta Onexik$. The apparently complete behavior of S_2 12 to move can be recognized as the familiar effect of interchanging the next links of In the edge algebra, the Org and $L_{O}f$ rings of an odge z are the orbit and rrow rest. For effect orbits under $Orst rol f c and c Orst rol f construction of a new edge algebra <math>A = [E, E', Orst', R_O, [Fig)$ from an artisting algebra $A = [E, E', Orst', R_O, [Fig)$ where Oratic f solution $Oration Oratic <math>A = [E, E', Orat, R_O]$, where Oratic f solution from Oratic A is by redefining some of its values. The modifications needed to obtain the effect described above are actually quite simple. If we let $\alpha =$ two circular list nodes [Kn].





The identities



As one may well expect to preserve the validity of the axioms F1-F3 and E1-E3 we may have to make some additional changes to the *Onex* function. For example, whenever, we redchine *eDnex* fu be some *edg* f. we must also redefine *ePip(Onex/1)* to be *FRip*, so equivalently, *FTipOnexI* to be *eTIp*, so. Splateo far, b) must perform at least the following changes in the function *Onexr*:

In a similar way we can prove F2. To conclude, let us prove F3, or $eFig/Onex/1^{-2} \neq e$ for all n. In other works, we have us about hat Fija plausys takes an Onex/ orbit to a different Onex/ orbit is lattices to show this for the orbits of elimeness of X; in fact the symmetry of Splice implies it is sufficient to show this for the orbit ofa.

3

(bOnext Flip) Onext' = a Flip (a Onext Flip) Onext' = b Flip $(\beta Onext Flip) Onext' = \alpha Flip$ $(\alpha Onext Flip) Onext' = \beta Flip$

(bOnext Flip) Onext

 $\alpha Onext' = \beta Onext$ $\beta Onext' = \alpha Onext$ bOnext = a Onext

a Onext' = b Onext

Let $aO_T = \{a_1 a_2 \dots a_{m-1} a_m(=a)\}$ be the orbit of a under the original Onext. The orbit of a Fip under Onext is then a Fip OT $T = \{a_m a_{m-1} \dots a_{j}^2 a_{j}^2\}$, where $a_j^2 = A_{FI}$ for all T. These woo orbits are disjoint: they cannot contain any of the edges a_i , β_i , $OnextFIP_{OT}$ or $\beta OnextFIP_{OT}$. one contains b if and only if the other contains bFlip. There are then only three cases to consider (see figure 5.3):

a Oner(f = bOnext, to satisfy atom ES we must have $a \in B$ iff bOnext = B, which is equivalent to $a \in B$ iff bOnext. We will take this as a precondition for the validity of Splites (a); the effect of this operation is not defined if a is a primal edge and bis duel to vice-result. Another problematic situation is when b =a Onext/Rp.

Note that these equations reduce to Onext' = Onext if b = a. Since

aOnetrifie Onetries = aFig. which contradicts F3. In this particular case, it is more convenient to define the effect of Spl.1ce[a, b] as being null, i.e. Onetri = Onetri. It turns out that, with only these two

exceptions, the equations above always define a valid edge algebra:



ğ

۽





229

Proof: Since Splice does not affect Flip and Rov, all axioms except F2. F3. E2 and F5 are automatically satisfied by A'.

Theorem 5.1. If A is an edge algebra, a and b are both primal or both dual, and $b \neq a$ others(Filp, then the algebra A' obtained by performing the operation Splice(a, b) on A is also an edge

Since a and b, are both primal or both dual, the same is true of α and β . a OnextFlip and bOnextFlip, and $\alpha OnextFlip$. Thus the assignments corresponding to Splice(a, b) will not desuro <u>ت</u>

Now let's show E2 holds in A', i.e. e Rot Onex' Rot Onex' = e. Let X be the set of edges whose Onext has been changed, i.e. $X = \{a, b, \alpha, \beta, a \text{ Onext Flip, b Onext Flip, } \alpha \text{ Onext Flip, } \beta \text{ Onext Flip} \}.$

First, if eRot $\not\in X$, then eRotOnextRot $\not\in X$ OnextRot = X, and so

= (e Roi Onexi Roi) Onexi e RotOnext' RotOnext' = e RotOnext RotOnext

نه اا

Now assume $eRot \in X$. Notice that Splice(a, b) does eractly the same thing as Splice(b, a), Splice(a, b), and Splice(Daurstript). Now without loss of generality we can assume eRot = a. Then

¹ Note that if a mod b for in distinct subsychers A_{a} and A_{b} of A_{i} then the union of A_{a} mod the dual of A_{b} is also a valid color slighten. So, in practice we can always perform Spillen (a, b) when a and b lie in disjoint data graduats

As an example of the use of Spiles, consider the operation SwapDiagonal[2] that we will be using later on: given an edge e whose fit and right faces are triangles, the problem is to defere and connect the other two vertices of the quadrilateral thus formed and connect the other two vertics (see fig 5.4). This is equivalent to

a ← e.Oprev; b ← e.Syma.Oprev;

Splice[e, a]; Splice[e.Sym, b]; Splice[e, a.Lnert]; Splice[e.Sym, b.Lnext]

an d The first pair of \$p11ces disconnects e from the edge structure, leaves it as the single edge of a separate spherical component. last two \$p11ces connect e again at the required position.



Figure 5.4. The effect of SwapEdge [e].

Theorem 5.1. An arbitrary subdivision P can be transformed into a collection of |EP| isolated edges by the application of at most 2|EP|Splice operations Proof. Let e be an arbitrary edge of P. Then the operations Splitee $[e_{i} e Opren]$ and Splitee $[e_{2})m_{i}$, $e_{2})mOpren]$ will have the effect of removing e from P and placing it as an isolated edge on a separate manifold homeomorphic to the sphere. By repeating this for expansion to the operation of the operation of the forevery edge the theorem follows

we can conclude that any simple subdivision P can be constructed in O(|EP|) time and space, by using only the Splice and WaxeEdge From this theorem and from the fact that Splice is its own inverse. operations.

be too capensive. However, even in such applications it is frequently the case that we can defer those updates until they are traily needed (so that their cost can be amortized over a large number of Spiltees), or initialize the Data inks right from the beginning with the values or restrict assignments to them. Usually each application imposes geometrical or other constraints on the Data fields that may be affected may change the geometrical parameters of a large number of edges and Since they carry no topological information, there is no need to forbid by changes in the topology. Some care is required when enforcing those constraints. for example, the operation of joining two vertices faces, and updating all the corresponding Data fields every time may The Data links are not affected by (and do not affect) the MakeEdge and Splite operations; if used at all, they can be set and updated, at any time after the edge is created, by plain assignment statements. they must have in the final structure

6. Topological operators for Delaunay diagrams

In the Voronoi/Delaunay algorithms described further on, all edge variables refer uo togets of the Delaunay diagram. The **basis** field for a Delaunay edge r points or a record containing the coordinates of is origin *eDgs*, which is one of the sites: accordingly, we will use lise origin *eDgs*.

e. Drg as a synonym of e. Data in those algorithms. For convenience, we will also use e. Dest instand of e. Syno. Torg. We combasize again that these Dest and Org fields carry no uppological meaning, and are not updated by the Splice operation per se. The endpoints of the dual edges (Voronoi vertices) are meiher computed nor used by the algorithms, if detired, they can be easily added to the structure, either during is construction of after it. The fields e. Not. Data and e. Rot.⁻¹. Data are not used.

The topological manipulations performed by the algorithms on the Delaunay/Voronoi diagrams can be reduced to a pair higher-to-eit topological operators, defined here in terms of Sh124e and MaxeEdge. The operation $e \leftarrow$ connected, b, size? will add a new cdge e connecting the destination of a to the origin of b, arcseither at *Lf* or *aligh*, depending on the size flags. For added convenients, it will also set the torg and base fields of the new cdge to a loss t b.Org, respectively.

- e ← MakeEdge[]:
- () Org a Dost; a. Dost b. Dost; a. Dost b. Org; b)lice(a. If Side=latt THEN a. Lmart ELSE a. Sym]; Splice(a. If Side=latt THEN b ELSE b. Oprev] Splice(a. Sym, IF Side=latt THEN b ELSE b. Oprev]

The operation DeleceEdge[e] will disconnect the edge e from the ress of the structure (hiki may cause the rest of the structure to fall apart in two separate components). In a sense, DelecteEdge is the inverse of Connect. It is equivalent to

Splice[e.Sym., e.Sym.Oprev] Splice[e, e.Oprev];

7. Voronoi and Delaunay - some basic facts

In this section we recapitulate some of the most important properties of Voronoi diagrams and their duals, with an emphasis on the results ams and their duals, with an emphasis on the results r on. For a fuller treatment of these topics the reader should consult [Sh], [Le], or [SH]. we will need later

If we are given only two sites, then the associated Vorthoi regions are simply the two (open) half planes delimited by the biostent of the two sites. More generally, when a sites as given, the region associated with a particular site will be the intersection of all half planes containing b and delimited by the bisectors between p and the other sites. It follows that the Vortioni regions are (possibly unbounded) convex polygons, whose edges are portions of inter-site bisectors, and whose vertices are circumcenters of triangles defined by three of the sites. An example Voronoi diagram for a small collection of sites is shown in fig. 7.1.

vertices are the n sites and whose edges are straight line segments wertices are the n sites and whose edges are straight line segments that connect every pair of sites associated with regions that share a data common edge. If no four of the sites happen to be coefficial, then this dual is in fact a triangulation and, in any case, additional edges can be introduced to make it one. Although our algorithms can handle the degeneracies implied by four cocircular points, in this section only we will assume that such degeneracies do not arise, so we can speak mambiguously of the Delaunay triangulation of the n sites. If Sunambiguously of the Delaurary triangulation of the n sites. If S denotes our collection of sites, then D(S) will denote the Delaurary triangulation of S. triangulation. As the name implics, this is a planar subdivision whose As mentioned above, most of the time we will actually deal with a dual of the Voronoi subdivision, commonly called the Delaunay

We say that a circle is *point-free* if none of the given sites is contained in its interior. It is known that the circumricited of each Delaunay transfer is point-free. Actually the converse is also true. If the circumrick of a transfer defined by three of the sites is point-free.

a point-free cirtle passing through two sites, then these sites will be connected by a Delaunay edge. For a discussion of these facts see [Le], We will speak of these circles as wintersus to the "Delaunayhood" of the corresponding triangle or edge respectively. For completeness we also state here a ouple of other obvious properties of the Delaunay triangulation which will be useful in the sequel. triangle is a Delaunay triangle. And if it is possible to find then that



Figure 7.1. The Voronoi diagram (solid) and the Delaunay diagram (dashed).

Lemma 7.1. Each triangulation of n points of which k lie on the convex hull has 2(n-1) - k triangles and 3(n-1) - k edges. **Proof:** Let t be the number of triangles and e the number of edges. By comming edges we have 3t + k = 2s, and from Euler's relation n - e + (t + 1) = 2. We can solve this system of two equations and obtain the lemma. \Box

Lemma 7.2. Each convex hull edge is a Delaunay edge.

Proof: The half-plane supporting the edge and not containing the convex hull is a degenerate circle witness to the Delaunayhood of the edge. 🗆 Lemma 7.3. If S and T are two point sets then each Delauray edge $e \in D(S \cup T)$, both of whose endpoints are in S, is also, a Delaurary edge in D(S). In other words, the addition of extra points does not add new edges

between old points.

Proof: There is a point-free circle through e in $S\bigcup T$, and therefore a fortiori in S.

The InCircle test - implications for Delaunay œ

We now proceed to define the main geometric primitive we will use for Detaunary computations. This test is applied to four distinct points in the plane A, B, C, and D. See fig. 8.1.



Figure 8.1. The InCircle test.

efinition 8.1. The predicate $InCircle(A_1, B, C, D)$ is defined to be true if and only if point D lies interior to the region of the plane bounded by the circle passing through A, B, and C and lying to the left of that circle when the latter is traversed in the direction from A to B to C. Definition 8.1.

the points A, B, and C define a counterclockwise oriented triangle, and output, if they there a lockwise oriented one. (In ease A, B, and O, and C are collinear we interpret the line sa circle by adding a point at infinity). If A, B, C, and D are coinvillat, then our predicate returns false. Notice that the test is equivalent to asking whether ABC + LCDA > ZBC + LDAB. Another equivalent form of LABC + LCDA > ZBC + LDAB. In particular this implies that D should be inside the circle ABC if it is given below, based on the coordinates of the points.

Lemma 8.1. The test InCircle(A, B, C, D) is equivalent to

° V			
-	•	-	-
$x_{A}^{2} + y_{A}^{2}$		z ² + 3 ²	z ² + v ³
5	e s	ŝ	JD
¥.	- 	с Н	Чр
D(A, B, C, D) =			

Proof: We consider the following mapping from points in the plane to points in space:

 $\lambda:(x,y)\mapsto(x,y,x^3+y^3),$

which lifts each point on the x, y-plane onto the paraboloid of revolution $z = x^2 + y^2$. See Eq. 2.1 for an illustration. We first show that A, B, C, and D are corricular if and only if $\lambda(A), \lambda(B), \lambda(C)$, and $\lambda(D)$ are cophanat, a rather amazing fact.



Figure 8.2. The quadratic map for computing InCircle.

degenerate case where they are collinear, then $\mathcal{P}(A, B, C, D)$ is zero, as we ease by repanding it by the indit obtume 1D(A, B, C, D)is also the elgendy volume of the terthadron defined by A(A), A(B), A(C) and A(D). Since the volume is zero, the points must be coplanar. Otherwise it (i, q) denote the center and τ the radius of the circle passing frongh the points A, B, C, D. We must have Suppose first that A, B, C, and D are cocircular. If we have the

$$(x_A - p)^2 + (y_A - q)^2 = r^2,$$

equivalently ъ

$-2p \cdot x_A - 2q \cdot y_A + 1 \cdot (x_A^2 + y_A^2) + (p^2 + q^2 - r^2) \cdot 1 = 0.$ (1)

This relation also holds for points B. C, and D, and therefore we have a linear dependence among the oblumns of the determinant $\mathcal{D}(A, B, C, D)$, which implies that its value is zero. So again we can conclude that $\lambda(A), \lambda(B), \lambda(C)$, and $\lambda(D)$ are coplanar.

Now conversely suppose that $\lambda(A)$, $\lambda(B)$, $\lambda(C)$, and $\lambda(D)$ are coplanar. If all of A, B, C, and D are collinear, then we are done. So suppose, without loss of generality, that A, B, and C are not collinear. As above, let (p,q) denote the center and r the radius of the circumcircle of triangle ABC. Then A, B, and C satisfy equation (1) above. Since A, B, and C are not collinear, the corresponding three rows of $\mathcal{P}(A, B, C, D)$ are linearly independent. But all four rows are linearly dependent, since the determinant is zero. So the last row can be expressed as a linear combination of the first three, and therefore point D satisfies (1) as well , i.e., it is on the circle ABC.

revolution $z = x^2 + y^2$ project onto circles in the x,y-plane. The paraboloid is a surface that is convex upwards, and therefore, in a section of it with a plane, the part below the plane projects to the The above result shows that planar sections of the paraboloid of the plane to the exterior. From this and the standard right-handed orientation convention for the sign of volumes the lemma follows. Notice that this establishes an interesting correspondance between interior of the corresponding circle in the x, y-plane, and the part above circular queries in the plane and half-space queries in 3-space.

As a side note we remark that Ptolemy's theorem in Euclidean plane geometry does not lead to a useful implementation of the InCircle test as we always have

$AB \times CD + BC \times AD \ge BD \times AC$,

with equality only when the four points are cocircular. In fact, the quantity one obtains by rendering $AB \times CD + BC \times AD - BD \times AC$ radical-free is essentially the square of the determinant D(A, B, C, D)

The InCircle test possesses several interesting properties:

Lemma 8.2. If A. B. C. D are any four non-cortexilar points in the plane, then transposing on objecting the in the predicate Infertiel(A, B, C, D) will change the walks of the predicate from Infertiel(A, B, C, D). Will change the walks of the predicate from the to folds. or versa. In particular, the boolons sequence infertiel(A, B, C, D). Infertiel(B, C, D, A). Interde(C, D, A). Infertiel(D, A, B, C) is either Trans. T. F. or F, T, F, T.

Proof: This is obvious from the properties of determinants.

and lnCircle(A, B, C, D) is true, then lnCircle(C, B, A, D) is false, so reversing the orientation flips the value of the predicate. Whenever The fast two lemmas show that in InCircle(A, B, C, D) all four points play a symmetric role, even though from the definition D seems to be special. If A, B, C, D define a counterclockwise oriented quadrilateral

we consider applying this test to a (rooted) quadrilateral in the sequel, we will take the quadrilateral to be counterclockwise oriented.

The second set of the second set of the set of the second set of It. We can either draw diagonal AC or diagonal BD. By herma-12 the taskes AB, BA, C. D3, and DA are always pleatumay udges. If $Dicrite(A_1, B_1, C_1)$ is false, then circle ABC is point-free, so the diagonal AC completes the Delaunay uriangulation for the four points A. B. C. P diests the regulated D does). More generally, for any edlestion of points, all the convex hull edges are Ubelauray edges. For each northull Delaunay edges BD. if the two triangles ABD and BDC each northull Delaunay edge BD. if the two triangles ABD and BDC For a convex quadrilateral ABCD there are two ways to triangulate

A useful intuition about the *InCircle* test comes from the following observation. Given two points in the plane X and Y, the set of retriefs passing through X and Y forms ao none parameter family. For example, we can think of these circles as the family (c_i) , where t line XY. Co denoted the circle with diameter XY, and C_{∞} denotes the corresponding to right hildshare of XY. Note that the protion of these circles to the left of XY strictly decreases (to proper inclusion) as 1 increases, while the portion to the right of XY strictly denotes the circle corresponding to the half-plane to the left of the ncreases. See fig. 8.3. The next lemma uses this intuition to derive mother property of the Delaunay triangulation that will be useful to denotes the signed distance of the center of such a circle along the bisector of XY, measured from the midpoint of XY. Thus \tilde{C}_{-} is in the next section.



oints.

emma 8.3. Let A be a site and consider any line l through A. For conventience of terminology: assume that l is horizontal. Let N.1, N....., denote the Delaunay neighbors of A hyug above L listed in counterclockwase acte. If X is any point of L to he right of A. let \Gamma, denote the portion of the circumvircle of AXN, that emma 8.3.

is above L. Then the sequence $\{\Gamma_i\}$ is unimodal, in the sense that there is some j, $1 \leq j \leq k$, such that for $1 \leq i < j$ we have

Proof: We first show that if X_i denotes the rightmost intersection of $\Gamma_i \supseteq \Gamma_{i+1}$, while for $j \leq i < k$ we have $\Gamma_i \subseteq \Gamma_{i+1}$.

the circumcircle of triangle AN_iN_{i+1} , i = 1, 2, ..., k - 1, with C_i then the sequence of points $\{X_i\}$ moves monotonically to the keft.

Consider, as in fig. 8.4, three consecutive Delaunay neighbors $N_{\rm c}$, $N_{\rm r+1}$, and $N_{\rm r+2}$ of $A_{\rm c}$. The point $N_{\rm r+2}$ is usuable the circle $M_{\rm r}N_{\rm r+1}$ and $N_{\rm r+2}$ of $A_{\rm r}$ and $N_{\rm r+2}$ are one opposite sides of $A_{\rm r+1}$. So to get to the circle $A_{\rm N+1}, N_{\rm r+2}$ from $A_{\rm N}N_{\rm r+1}$, while always passing through and $M_{\rm r+1}$, we must expand on the side of $N_{\rm r+2}$, and therefore we must contract on the side of $N_{\rm r+1}$, and $M_{\rm r+1}$ in this powers our ascention.



of X, then X is inside the current $\sim \infty_{MS}$ as use α_{MN} , α_{MN} are used and α_{MN} . M_{k-1} is inside the currentrice of $AN_{k}X$. After the X, move to the left of $AN_{k}X$ and $M_{k}X$. Thus the Left of $AN_{k}X$. Thus the Left of $AN_{k}X$. Thus the Left of $AN_{k}X$. Thus the To prove the lemma now, note that as long as the X_i are to the right

We will refer to lemma 8.3 as the *unimodality lemma*. Another useful property of the Delamay triangulation that can be derived with the help of the *InCircle* test is:

Lemma 8.4. Let A be a site and let X and Y denote any two distinct Delaunay neighbors of A. that lie inside the corvex angle $\mathcal{L}XAY$ are inside the circumcircle of triangle AXY, and all neighbors lying outside the angle are outside that same circle.

 $\mathcal{L}XAY$ the ray AY is counterfectives of AX, Suppose the lemma is faits mission the converse angle $\mathcal{L}XAY$. Let N be the Delaunay neighbor of A outside the converse angle $\mathcal{L}XY$ which is the fast to be cocountered as we seep this angle in counterfockwase fashion. Let also M denote the predecessor of N, as in fig. 8.5. Note that N is also outside the circle AMN since X, M_A and N are all on the same side of AY. Therefore Y is single the contraction Delaunayhood A similar argument proves the lemma outside $\mathcal{L}XAY$. Proof: Without loss of generality let us assume that in the convex angle



Figure 8.5. A property of the Delaunay diagram.

The divide and conquer algorithm.

are treated, into up sourcest primer provides primer, some variance, and the predicate CCW(A, B, C), which is true if the points A, B and C form a counterclockwise oriented infangle. The latter is a standard on in geometric algorithms and is equivalent to a reporting on which side of a line a point liss. It is equivalent to increding A, B, G, D, for D chosen as the barycenter of triangle ABG. tancously computing the Voronoi diagram as well. Our algorithm is the dual of that proposed by Sharnos and Hocy [SH] and like theirs. Turns in time $O[n \, g \, n]$ and uses linear storage. In [Le] D. T. Lee proposes a dual algorithm which is similar to but more complex than ours. Unlike both of these, our algorithm need not compute analyze, and prove correct a divide and conquer algorithm for computing the Delaunay triangulation of π points in the plane. Topologically the quadrodge data structure gives us the dual for free, so by associatstraight line intersections unless the coordinates of Voronoi vertices are needed. The only geometric primitives used by our algorithm are In this section we use the tools we have developed so far to describe, ing some relevant geometric information with our face nodes, e.g. the coordinates of the corresponding Voronoi vertices, we are simul-

partitioning our points into two halves, the fet half (L) and the right half (R), which are separated in the z-coordinate. We next recursively compute the Delaunay triangulation of each half. Finally we need to marry the two half triangulations into the Delaunay triangulation of the whole sec. As one might expect, in the divide and conquer algorithm we start by

We now elaborate on this brief description in stages. First of all it is advantageous to start out by sorting our points in increasing a coordinate. When there are ties we resolve them by sorting in Sin L makes all future splittings constant time operations. After splitting in the middle and recursively triangulating L and R, we must consider the merge step. Note that this may involve deleting some L-L or R-R edges and will certainly require adding some L-R or crossedges. By lemma 7.3, however, no new L-L or R-R edges will increasing y-coordinate and throwing away duplicate points. be added.

What is the structure of the cross edges? All these edges must cross a line parallel to the y axis and placed as the splituing x value. This establishes a linear ordering of the cross edges. So we can talk about successive cross edges. the bottom-most cross edge, ec. The algorithm we are about to present will produce the cross edges incrementally, in accending y-order. See fig. 9.1.



Figure 9.1. The structure of the L-R edges.

neighbors of A in L that are above basel. We prove our kernna inductively on the sequence of cross edges discovered. (ValidR[rcand] AND InCircle[lcand.Dest, lcand.Drg, rcand.Drg, rcand.Dest]) if both are valid, then choose the appropriate one using the InCircle uss. } if leared is the winner then make the next cross edge close the triangle with *keand* and bazel, } else make the next cross edge close the triangle with reand and bazel } (let ValidL and ValidR be procedures which test whether *leand* and reand are above base!) { advance kand counterfockwise, deleing the old kand ddge, unuil the InCircle test fails } IF ValidL(leand) AND ValidL[leand.Dnest] THEN WHILE InCircle[leand.Dnest, leand.Onest, leand.Org, basel.Org] DO initialize leand to the counterclockwise first edge incident to basel Dest and above basel } (reate a first cross edge (bursh) by computing the lower common tangent of L and R } 20 WHILE CCF[1d1.0rg, 1d1.0rg, 1d1.0rg] D0 1d1 + 1d1.1meat 0D; IF CCF[rd1.0est, rd1.0rg, 1d1.0rg] THEM rd1 + rd1.Rprev ELSE EXIT; 0D; if both *leand* and *reand* are invalid then *basel* is the upper common tangent } F NOT ValidL[Leand] AND NOT ValidR[reand] THEN EXIT; #alidL[1] : CCW[basel.Org, basel.Dest, I.Dest]; #alidR[r] : CCW[basel.Org, basel.Dest, r.Dest]; basel ← Connect[rcand, basel.Sym, left]; rcand ← basel.Sym.Lnert; advance base/ and rcinitialize leand or reand } iscration to the edge being deleted. Similarly, iterations of the recard loop can be charged to deleted edges. The rest of the body of the main loop requires constant time and may be charged to the L - Lor R - R edge closing the next transfit. This shows that the overall cost of the merge pass is linear in the size of L and R. We now elaborate on the above program. Recall first that the number of vertices, edges, and faces in a triangulation are all linearly related. Also, the lower common langent computation takes linear time as either ldi or rdi advances at each step. What about the cost of the leand computation? We can account for this loop by charging every symmetrically, advance reard clockwise } THEN BECIN { connect to the right } ELSE . . . { connect to the left } connect to either lcand or rcand } [F NOT ValidL[lcand] OR { this is the merge iteration } DeleteEdge[lcand]; symmetrically for rcand t + lcand.Onext; lcand + t; OD: ENG 8 6 8 We can intuitively view what happens by imagining that a circular bubb waves its way in the space teverse 1.5 and R and in so doing gives us the cross cdgss. Inductively we have a point free circles doing gives us the cross cdgss. Inductively we have a point free circles action. Consider continuously maniforming his circle into other circles avoing Azerl as a chord, but lying further into the half plane above basel. As we remarked, there is only a single degree of freedom, as the basel. As we remarked, there is only a single degree of freedom, as the edge from the right end-point of base! to one of the L-neighbors of the left end-point lying: "bowe" base! in the program below edges from the full end-point of base! to its candidate L-neighbors will be had in the variable keand, and their symmetric counterparts in *reaul.* See fig. 9.2. the first L point to be encountered in this process and coordinations. R point. A thail area chooses the point among these we that would be encountered first. We start the tross edge intention by computing the lower common angent of L and R, which defines the first encountered the edge first is the clockwise convex hull edge out of the fightmost L point, and first the counterlickwise convex hull edge out of the point, and first is the contractickwise convex hull edge out of the L point, and first is the counterlickwise convex hull edge out of the L point, and first is the counterlickwise convex hull edge out of the convext and first is the counterlickwise convext hull edge out of the lower and first is the counterlickwise convext hull edge out of the lower and first is the counterlickwise convext hull edge out of the lower and lower l Influences R point. We assume these are returned by the recursive calls of the triangulation procedure to L and R. In the program below proces in $\{I\}$ is comments, certain portions of code that are just symmetric variants of others shown are elided and are indicated by center of the circle is constrained to lie on the biscoro of basel. See fig. 9.3. Our circles will be point free for a while, but unless basel is the to L or R, giving rise to a new triangle with a point-free circumcircle. The new L - R edge of this triangle is the next cross edge determined L-R edge of this triangle is the next cross edge determined upper common tangent of L and R, at some point the circumference of our transforming circle will encounter a new point, belonging either In more detail, edge leand is is computed so as to have as destination by the body of the main loop below.

Lemma 9.1. Any two cross edges adjacent in the y-ordering share a common vertex. The third side of the triangle they define is either an Proof: Any two consecutive intersections of a transpulation with a straight line must belong to the same transpulation were. Therefore the straight line must belong to the same transpulation, and the wind side of the margic is fully to one side of the order of the vertical bind side of the margic is fully to one side of the order of the vertical to the vertical section.

common veriex. The third s L - L or an R - R edge.

Lemma 9.1 has the following important consequence. Let us call the current cross edge the base, and write its directed variant going from right us that as basel. The successor to basel will eithere be an edge going from the first encipoint of *basel* to one of the *R*-netghbors of the right end-point lying "above" basel, or symmetrically, it will be an

divider. 🗆



Figure 9.2. The variables lcand, rcand, and basel.









We now formally state the lemmas that prove the correctness of the above algorithm. Lemma 9.2. When the leard iteration stops, the circumcirtle of the tri-angle defined by leard and basel is free of other L points. Furthermore all deleted edges are not in the final Delauray triangulation.

Proof: Let A and X denote respectively the left and right endpoints of base! and, as in lemma 8.3, let N_i , $1 \le i \le k$, denote the Delaunay

Figure 9.3. The rising bubble.

circle upwards while always making it pass through A and X. At some point it will encounter a new L point N for the first time. This point will be above base! We wish to show that N is the point that the kand circumcircle of the triangle defined by bazel and the previous cross edge is the circumcircle of a Delaunay triangle and therefore it is point-free. As in our earlier discussion, suppose that we "push" this iteration will discover. To see this note first of all that N must be a Delaunay neighbor of A, since our circle is a Delaunayhood witness. The *leand* iteration proceeds until the first time X fails outside the Assuming that the algorithm has worked correctly up to now, then the circle AN_{v+1} , as shown in the code. By the unimodality lemma, this gives us the neighbor $N = N_v$ which determines the circle AXN_v with the smallest extent above basel. This circle now forms the basis for the next iteration through the main loop.

The inclusion of X in the circle AN_mN_m+1 for $m=1,2,\ldots,m-1$ establishes that AN_m is not a Icelannay edge in the final triangulation.

since each circle with AN_m as chord contains either N_{m+1} or X in its intercerve. Thus the deep decisions performed are valid. It is easy to check that the cross edges drawn, together with the remaining edges of L and R define a triangulation of the n sites. This is exentially the proof, madulo capefuling arms care on cratan boundary cases. We must first of all show that the *leant* freation, if started at all never gets to neighbors on the bottom side of bazet. For if the retration was invoked, then there must be at least two regibbors above bazet. We claim that the moment *leand* Omer crosses to the bottom of bazet, the WHILE *ln/livel* test fails. See **fig. 34**.



Figure 9.4. End of 1cand loop.

In this case we had that X was inside the circle $AM_{n-1}N_{k-1}$ If the angle $CM_{k-1}N_{k-1}$, is obviously on toxic the circle $AM_{k-1}N_{k-1}$, so the *InCircle* test fails. If the angle $ZM_{k}M_{k+1}$ is concerve the N_{k} is obviously the coreave, then we obtings happen. First of all the circle $AM_{k}M_{k+1}$ is concare, then we obtings happen. First of all the circle $M_{k}M_{k+1}$ is concare, then we obtings happen. First of $M_{k}N_{k-1}N_{k}$ is outside the circle $AM_{k-1}N_{k-1}N_{k-1}N_{k}$ but on the same side of AM_{k} as N_{k-1} , and X. X now fails inside the circle $AM_{k-1}N_{k}$ but on the same side of AM_{k} as N_{k-1} , and X. X now fails fails inside the circle $AM_{k-1}N_{k+1}$. So in this case the *InCircle* test fails also.

Finally note that the main loop starts with our accending circle being the halphane below the lower outer common tangent of L and R, and ends with that circle being the halphane above the upper such angent. \Box

A symmetric statement holds for the *read* iteration. Furthermore, the final choice between *keand* and *reand* is again done by the *luclinic* test, namely *reand* will be chosen if *lacterel*[cand.Dest, leand.Org, reand.Org, reand.Dest] is true. It now follows that: Lemma 9.3. The triangle defined by basel and the next cross edge as computed by the above algorithm that a circumcitle free of both L and R points, and thus is a Delaunay triangle. Furthemore, after the completion of the cross edge iteration, all remaining L or R transfer are also Delaunay triangles for the full configuration. Proof: The first statement is obvious from the above discussion. The eccond statement is most easily proven by showing batte each remaining L = -L or R = -R edge is stationary under the swapping rule. The only possible supercledges are those bodtering trangits containing cross edges. It follows that these edges were *leand* or *reard* at some point in the algorithm and were not detect. So the *InCircle* test of the rule short in the swapping rule. The rule is a rule in the reard the swapping rule.

These lemmas complete the proof that the above algorithm correctly computes the Delaunay triangulation. The algorithm will work with

cocircular sites and other degenerate cases. When both *leaved and reand* are equally good, it advirarity leaves reard. In practice floating point ternes in the computation of the *InCircle* etta usually interfere with any effort to handle degenerate cases in a consistent way.

It is worthwhile to comment on a way to view his algorithm as operating in three-dimensional space, on the filted images of our sites under the map λ discussed in the proof of lemma 81. Three lithed images are points on a convex statister, and therefore they define a convex polyhedron, corresponding to their convex hull. The discussion in the proof of lemma 81. It as also established that the "downwards" looking faces of this polyhedron are in a one-uous correspondance with the Delaunsy faces of the sites. The upwards facing faces of the polyhedron correspond to transfels of our collection of the polyhedron correspond to transfels of our collection of sites. Note that when our cores define and RS. The upwards facing faces of the fullyhedron correspond to transfels of our collection of sites. Note that when our cores define the sites. If we now ket this ferate contract unal it his the first site, while always having as chord the land of the furthest point. Voronoi, In fact, If our cross degi terchon is the loc containing all the sites. If we now ket this ferate constatue, but by using the furthest point versions of L and R at diversing the scarse disk of the sites of were common under the advection protection of the March fer were site and the constatue to the state of the larch fer site were common under the advection of the furthest point versions of L and R at diversing the scarse disk to the lower common under of L and R.

From the above discussion it is appretu that our divide and conquer algorithm is computing the convex hull of the lifed images of the sites. It is in flact exactly the Preparata Hong [PH] algorithm for computing the convex hull of points in three dimensions. If the h/r/r/ret was is replaced by a "megative volume" text, as obtained by substituting in Expendent by a "megative volume" text, as obtained by substituting in the replaced by a "megative volume" text, as obtained by substituting in the replaced by a "megative volume" text, as obtained by substituting in the replaced by a "megative volume" text, as obtained by substituting in the replaced by a "text" of the relation o

10. Incremental techniques

The algorithm of the previous section assumes that all points are known at the beginning of time. For many spplications are are interested in dynamic algorithms, which allow us to update our diagrams as new points are added or deteted. The machinery we have developed so far yields also simple and elegant insertion algorithms. Deletions present problems.

It will simplify our discussion to assume that our points are always within some large concre polygon (sy a trianglo) whose writtes are considered to be among the given points. Thus the entire interior of this polygon will have been triangulated, and when a new point is given, we can be sure that it will fall within one of the eristing triangular faces. The algorithm for updating the Delamay tirangulation are not by locating the new point in one of the triangular faces of the subdivision. This can be done by Kirkpatrick's method in time O(gan) [Ki1]. Our concern is with how the triangulation is to be updated.

comma 10.1. The edges connecting the new paint to vertices of its enclosing triangle are Delaunay edges.

Proof: From fig. 10.1 it is clear the we can always find a circle contained inside the circumcircle of a triangle ABC, and passing

through one of A, B, or C, and a specified interior point X of that circle. Such a circle is point-free. \Box



XC.

.

However, the edges of the enclosing triangle are themselves suspect. We can remove suspect edges by the swapping tale discussed in section 8. We This gives rise to a very simple algorithm, given in high-level form, below.

place the three edges of the enclosing triangle on a stack in ccw order;

WHILE the stack is nonempty DO

BEGIN

{ each edge on the stack forms a triangle with the new point } remove an edge from the stack; consider the quadrilateral whose disconal

consider the quadrilateral whose diagonal is this edge;

if this edge passes the swapping rule test, then forget about it,

else apply the swapping rule and stack, in cow order, the two edges not

incident to the new point; OD; At any one time the set of triangles with the new point X as a verter form a survised polygon round X. This is because the polygon can only get larger with the early polygon the way rule on a bounding edge, and if it does grow, then the two new stacked sides are in the negative screen with verter X and definited by the deleted deg. Furthermore, because we stack edges in a consistent order, but are polygon section of the perimeter of the polygon, with the rest of the perimeter of stack polygon, with the rest of the perimeter of stack polygon.

Lenuma 10.2. All edges incident to the new point introduced during applications of the swap rule in the above algorithm are Delaunay edges. So is each forgotten edge Proof. Let X be the new point and suppose that the swap rule was just applied to quadrilateral XLMN where it represed diagonal LN with diagonal XM, as in fig. 10.2. Then X must be interior to the circle LMN and by the same argument as in the proof of Remma 10.1 it follows that XM is a Delamay edge. If the swap rule thield D it follows that XM is a Delamay edge. IN the swap rule the Delamay and can be forgotten. \Box

Lenuma 10.3. No edge is ever stacked twice.

Proof: One of two things can happen at each application of the swapping rule. If the swap succeeds, then the polygon grows past the old bounding edge, so that edge can never be on its boundary again.

If it fails, then that bounding edge is frozen and the algorithm will never again touch any edge in the angular sector with vertex X and delimited by that edge. \Box



Lemma 10.4. When the above loop terminates the swap test fails everywhere in the triangulation.

Proof. From lemma 10.2 we know that all edges leading to the new point X and introduced during the above algorithm are Debunary degas. The only other edges bunding new triangles are the edges of the star-shaped polygon discussed in the proof of lemma 10.3. At the termination of the loop the stack is empty and so all the bounding edges have been forgotten at some point. Thus there are no suspect edges have Deta.

The above lemmas imply that this loop correctly computes the final Delayor transgulation, and that it to does so in a most O(r) time. More proceedy if it edges meet to be added or deleted to update the Delauora structure. Unchost our algorithm works in time O(k) since each dedictly the algorithm must be added, and each edge that is deleted must be deteed.

We saw that the forgotten edges will always define a path around the new point, which will recruitable (see). If is an edge on the path, then the next edge is *e Lext Opre*. This observation gives us a way to implement the above algorithm without any autilary scorage, such as the stack by a method that is similar to that used in the provious section. Here keer stars out farbitrarily) as one of the edges connecting the new point to one of the vertices of its choice strange. By conventions we that of the nex point a being the field range is proventions we will always be invalid (noncristent, in fact) and we will consider for reand the various edges out of the forgotten edges of the previous algorithm. Note, however, that the forgotten edges of the previous section would connect all correspond to the forgotten edges of the previous section would connect all cores edges in strict countercleckwise onder.

Something quite general is happening here. We have a method with which, given may wo Delumay transluadions. Land R (non oressarity linearly separated), and a cross cage between them, we are able to find the next cross cage (if one exists) on a specified side of the original one. Thus we have another way to look at Kirkparick's [Kr3] linear into merge of two arbitrary Voronoi subdivisions. The above arguments show that it is possible to insert a new site into the Delaunay structure in total time O(k), if k updates need to be made. Unfortunately we know of no O(k) algorithm for handling the deletion of a site that leaves an untriangulated face of k sides. Our best algorithm has asymptotic complexity $O(k \lg k)$, which in the worst case $k = \Theta(n)$ is as bad as rebuilding the subdivision from scratch. We do not know of a linear algorithm even if we assume that the deletion of the site leaves a convex face. We regard the handling of deletions as the major open problem in this area.

11. Conclusions.

In this paper we have presented a new data structure for planar subdivisions which simultaneously represents the subdivision, its dual, and its mirror image. Our quad edge structure is both general (it works for subdivisions on any two-dimensional manifold) and space efficient. We have shown that two topological operations, both simple to implement, suffice to build and dismantle any such structure.

We have also shown how, by using the quad edge structure and the *InCircle* primitive, we can get compact and efficient Voronoi/Delaunay algorithms. The *InCircle* test is shown to be of value both for implementing and reasoning about such algorithms. The code for these algorithms is sufficiently simple that we have practically given all of it in this paper.

12. References.

- [Ba] Baumgart, B. G. A Polyhedron Representation for Computer Vision. AFIPS Conference Proceedings, Vol. 44, 1975 National Computer Conference, pp. 589–596.
- [Br] Braid, I. C., Hillyard, R. C., and Stroud, I. A. Stepwise construction of Polyhedra in Geometric Modeling; in Brodlie, K. W. (ed.), Mathematical Methods in Computer Graphics and Design. Academic Press, London, 1980, pp. 123–141..
- [Ev] Even, S. Algorithmic Combinatorics. Macmillan, N. Y., 1973.
- [GS] Green, P. J., and Sibson, R. Computing Dirichlet Tesselation in the Plane. Computer Journal, Vol. 21, no. 2, 1977, pp. 168-173..
- [Har] Harary, F. Graph Theory. Addison-Wesley, 1972, p. 105..
- [IK] Iyanaga, S., and Kawada, Y. Encyclopedic Dictionary of Mathematics (2nd. ed.). MIT Press, Cambridge, Mass., 1968..
- [Ki1] Kirkpatrick, D. Optimal Search in Planar Subdivisions. Technical Report 81-13, Department of Computer Science, University of British Columbia, 1981..
- [Ki2] Kirkpatrick, D. Efficient Computation of Continuous Skeletons. Proceedings of the 20th Annual Symposium on Foundations of Computer Science, 1979, pp. 18–27.
- [Kn] Knuth, D. E. The Art of Computer Programming, Vol. I: Fundamental Algorithms (2nd. ed.). Addison-Wesley, 1975..
- [Le] Lee, D. T. Proximity and Reachability in the Plane. Report R-831, Coordinated Science Laboratory, University of Illinois, Urbana, 1978..
- [MS] Mantyla, M. J., and Sulonen, R. GWB: a Solid Modeler with Euler Operators. IEEE Computer Graphics and Applications, Vol. 2 N. 5 (September 1982) pp. 17–31.

- [MP] Muller, D. E., and Preparata, F. P. Finding the Intersection of Two Convex Polyhedra. Theoretical Computer Science Vol. 7, 1978, pp. 217–236.
- [PH] Preparata, F. P., and Hong, S. J. Convex Hulls of Finite Sets of Points in Two and Three Dimensions. CACM, Vol. 20, 1977, pp. 87-93.
- [Sh] Shamos, M. I. Computational Geometry. Ph. D. dissertation, Yale University, 1977.
- [SH] Shamos, M. I., and and Hoey, D. Closest-Point Problems. 16th Annual Symposium on Foundations of Computer Science, 1975, pp. 151–162..